Author's response to reviews

Title: Treatment of Enterohemorrhagic E. coli (EHEC) infection and Hemolytic Uremic Syndrome (HUS)

Authors:

Paul N Goldwater (paul.goldwater@health.sa.gov.au)
Karl A Bettelheim (bettelheim@talktalk.net)

Version: 5 Date: 7 October 2011

Author's response to reviews:

I have added a few words based on a paper published today by Hickey CA, Beattie TJ, Cowieson J. et al. (Early volume expansion during diarrhea and relative nephroprotection during subsequent hemolytic uremic syndrome. Arch Pediatr Adolesc Med 2011; 165: 8840.) as this is highly relevant to the main theme of my paper.